Author's response to reviews

Title: Pain, psychological distress and health-related quality of life at baseline and 3 months after radical prostatectomy

Authors:

Kerstin Wickstrom Ene (kerstin.wickstrom.ene@vgregion.se)
Gunnar Nordberg (gunnar.nordberg@medfak.gu.se)
Fannie Gaston Johansson (fannie@son.jhmi.edu)
Bjorn Sjostrom (bjorn.sjostrom@his.se)

Version: 4 Date: 31 October 2006

Author's response to reviews:

Cover letter for the revised article:
Pain, psychological distress and health-related quality of life at base-line and three months after radical prostatectomy.

Qualifikations
The authors' qualifications have been removed

Abstract
The abstract has been included in the article

Conflicts of interest
Changed to competing interests

Figures
The figure number has been put before the figure title.
The figure has been cropped

Tables
Tables are included in the manuscript and removed from the Additional files